Author’s response to reviews

Title: Social Media Use by Physicians: A Qualitative Study of the New Frontier of Medicine

Authors:
Lauren Campbell (Lscampbell@challiance.org)
Yolanda Evans (Yolanda.evans@seattlechildrens.org)
Megan Pumper (Megan.pumper@seattlechildrens.org)
Megan Moreno (megan.moreno@seattlechildrens.org)

Version: 2 Date: 04 Jun 2016

Author’s response to reviews:

-------------------Editorial Requests-------------------

Please note that all submissions to BMC Medical Informatics and Decision Making must comply with our editorial policies. Please read the following information and revise your manuscript as necessary. If your manuscript does not adhere to our editorial requirements this will cause a delay whilst the issue is addressed. Failure to adhere to our policies may result in rejection of your manuscript.

Ethics:

If your study involves humans, human data or animals, then your article should contain an ethics statement which includes the name of the committee that approved your study.

If ethics was not required for your study, then this should be clearly stated and a rationale provided.

We have included the sentence: This project was considered exempt by the University of Washington Human Subjects Review Committee (#46467).

Consent:

If your article is a prospective study involving human participants then your article should include a statement detailing consent for participation.

If individual clinical data is presented in your article, then you must clarify whether consent for publication of these data was obtained.
We have included the sentence: We obtained verbal consent of each healthcare provider who participated.

Availability of supporting data:

BioMed Central strongly encourages all data sets on which the conclusions of the paper rely be either deposited in publicly available repositories (where available and appropriate) or presented in the main papers or additional supporting files, in machine-readable format whenever possible. Authors must include an Availability of Data and Materials section in their article detailing where the data supporting their findings can be found. The Accession Numbers of any nucleic acid sequences, protein sequences or atomic coordinates cited in the manuscript must be provided and include the corresponding database name.

We have added this section as follows:

Availability of Data and Materials

Data is available in the University of Washington Thesis Library database or upon request of the primary investigator.

Authors Contributions:

Your 'Authors Contributions' section must detail the individual contribution for each individual author listed on your manuscript.

We have a section describing this and have revised this to be by individual author as follows:

Author’s Contributions:

Lauren Campbell: LC was involved in all stages of this project. LC designed and conceptualized the study, conducted data collection, participated in analysis, drafted and revised the manuscript, and approved the final version of this manuscript.

Megan Pumper was involved in conceptualizing the study, participated in analysis, drafted and revised the manuscript, participated in key editing and approved the final version of this manuscript.

Yolanda Evans was involved in interpretation of analysis findings, drafted and revised the manuscript, participated in key editing and approved the final version of this manuscript.
Megan Moreno was involved in all stages of this project. LC and MM designed and conceptualized the study, participated in analysis, drafted and revised the manuscript, participated in key editing and approved the final version of this manuscript.